Author’s response to reviews

Title: In vitro antibacterial and antibiotic modifying activity of crude extract, fractions and 3',4',7-trihydroxyflavone from Myristica fragrans Houtt against MDR Gram-negative enteric bacteria.

Authors:

Joachim Dzotam (kamgue_joachim@yahoo.fr)
Ingrid Simo (simoingrid@yahoo.fr)
Gabin Bitchagno (gabin1256@gmail.com)
Ilhami Celik (celik@gmail.com)
Louis Sandjo (plsadjo@yahoo.fr)
Pierre Tane (ptane19@gmail.com)
Victor Kuete (kuetevictor@yahoo.fr)

Version: 4 Date: 05 Dec 2017

Author’s response to reviews:

Please make the following revisions to your manuscript:

1. We had noticed that your Acknowledgement statement contradicted your Funding statement in terms of funding. Please clarify, and move all funding information to the Funding section.

Answer: thanks very much. This has been done.

2. Please revise your Competing Interests section to clearly state you are on the editorial board for BMC Complimentary and Alternative Medicine. Please include your position title in this section.

Answer: thanks very much. This has been done